Author’s response to reviews

Title: Use Case Driven Evaluation of Open Databases for Pediatric Cancer Research

Authors:

Fleur Jeanquartier (f.jeanquartier@hci-kdd.org; fleur.jeanquartier@tugraz.at)
Claire Jean-Quartier (c.jeanquartier@hci-kdd.org)
Andreas Holzinger (a.holzinger@hci-kdd.org)

Version: 2 Date: 16 Nov 2018

Author’s response to reviews:

Dear reviewer #1,

First of all thank you again for your valuable comments.

We now extended the description of different characteristics between adult and pediatric cancers among adults vs. pediatric cancers as suggested.

Thank you also again for pointing out the ambiguity of our statement. We now do not refer to the term “incidence” but the “number of cases”. Since the incidence of a few pediatric cancer diseases can be high, there are still a few number of cases only due to the limited range of age for the particular disease subtype.

We added missing citations as recommended and also corrected some citation formatting issues.

Furthermore, we included your recommended specifications regarding the term “information”.

We described the use case scenario in more detail, and added suggested introductory sentences.

Thank you also for the general comments regarding editing style. Proofreaders further revised our manuscript.

We also corrected mentioned typos, changed the use case enumeration accordingly and further corrected a wrongly placed latex command.

We also revised the manuscript according to consistent quotation marks.

Last but not least we marked all changed paragraphs in red for better highlighting.
Dear reviewer #2,

Thank you again.